Author's response to reviews

Title: Validity of self-reported weight, height and resultant body mass index in Chinese adolescents and factors associated with errors in self-reports

Authors:

Xiaoyan Zhou (larkluna@163.com)
Michael J Dibley (mdibley@health.usyd.edu.au)
Yue Cheng (luckyyue@gmail.com)
Xue Ouyang (mnoyx@163.com)
Hong Yan (xjtu_yh.paper@yahoo.com.cn)

Version: 4 Date: 30 March 2010

Author's response to reviews: see over
Dear Editor,

RE: MS: 5298685033062856 - Validity of self-reported weight, height and resultant body mass index in Chinese adolescents and factors associated with errors in self-reports

We are grateful to you for your comments on our manuscript. We have revised our manuscript accordingly.

The formatting changes are as below:
We have reconstructed Table 4 and re-orientated it, and all the tables now fit into an A4 page (smaller than 176mm x 230mm) in portrait format. And we have removed all the line numbers.

Thank you for considering our revised manuscript.

Yours sincerely,

Hong Yan on behalf of authors

Professor in Epidemiology and Health Statistics
Dean of the College of Medicine
Xi’an Jiaotong University College of Medicine
P.O. Box, 46
No.76 West Yanta Road
South Suburb of Xi’an
Xi’an, 710061
Shaanxi Province
P.R. China